Author's response to reviews

Title: Effects of endotoxin exposure on childhood asthma risk are modified by a genetic polymorphism in ACAA1

Authors:

Joanne E Sordillo (rejoa@channing.harvard.edu)
Sunita Sharma (resss@channing.harvard.edu)
Audrey Poon (audreypoon@hotmail.com)
Jessica Lasky-Su (jessica.a.su@gmail.com)
Kathleen Belanger (kathleen.belanger@yale.edu)
Donald K Milton (dmilton@umd.edu)
Michael B Bracken (michael.bracken@yale.edu)
Elizabeth W Triche (elizabeth_triche@brown.edu)
Brian P Leaderer (brian.leaderer@yale.edu)
Diane R Gold (redrg@channing.harvard.edu)
Augusto A Litonjua (reaal@channing.harvard.edu)

Version: 8 Date: 1 December 2011

Author's response to reviews:

Parenthesis added on pg. 6.

On pg. 4 (last sentence in Boston Home Allergens paragraph) "patient" changed to "subject"